Author’s response to reviews

Title: Attitudes, Beliefs, and Perceptions of Caregivers and Rehabilitation Providers about Disabled Children’s Sleep Health: a Qualitative Study

Authors:

Xiaoli Chen (xchen@hsph.harvard.edu)
Bizu Gelaye (bgelaye@hsph.harvard.edu)
Juan Carlos Velez (jcvelez2002@yahoo.com)
Micah Pepper (micahpepper@gmail.com)
Sara Gorman (saragorm@gmail.com)
Clarita Barbosa (cibj71@yahoo.es)
Ross D Zafonte (RZAFONTE@mgh.harvard.edu)
Susan Redline (sredline1@rics.bwh.harvard.edu)
Michelle A Williams (mawilliams@hsph.harvard.edu)

Version: 2
Date: 22 July 2014

Author’s response to reviews:

1. Requesting for Ethics

--Response: We have revised our manuscript as below (page 6, line 72-74):

This study was approved by the institutional review boards of Centro de Rehabilitacion Club de Leones Cruz del Sur (IRB # 016) in Punta Arenas, Chile and Harvard School of Public Health (IRB # 22797-101) in Boston, USA. All participants provided written informed consent.

2. Requesting for Line Numbering

- -Response: We have added the line numbers in the text in the revised manuscript.